Author's response to reviews

Title: A web-based simulation of a longitudinal clinic used in a 4-week ambulatory rotation: a cohort study

Authors:

Rene WG Wong (Rene.Wong@uhn.on.ca)
Heather Lochnan (hlochnan@ottawahospital.on.ca)

Version: 6 Date: 28 January 2009

Author's response to reviews:

January 28, 2009
BMC Medical Education
Attention: Editors
To Whom It May Concern:

Re: MS: 1640056089221574

A web-based simulation of a longitudinal clinic used in a 4-week ambulatory rotation: a cohort study

Authors: Rene WG Wong and Heather Lochnan

This most recent revision of the manuscript has incorporated the formatting changes required prior to publication.

If there are any questions about this report, please do not hesitate to contact me.

Sincerely yours,

Rene Wong, MD, FRCPC